Author's response to reviews

Title: Life style related to blood pressure and body weight in adolescence: Cross sectional data from the Young-HUNT study, Norway.

Authors:

  Magnus H Fasting (magnusf@stud.ntnu.no)
  Tom IL Nilsen (tom.i.nilsen@ntnu.no)
  Turid L Holmen (turid.lingaas.holmen@ntnu.no)
  Torstein Vik (torstein.vik@ntnu.no)

Version: 4 Date: 4 April 2008

Author's response to reviews: see over
MS: 1589590217148604

Life style related to blood pressure and body weight in adolescence: Cross sectional data from the Young-HUNT study, Norway.

Dear editorial production team,

We have updated the list of authors and formatted the tables so that they fit on an A4 portrait page.

On behalf of the authors,
Yours, sincerely

Torstein Vik
Professor of paediatrics, MD, PhD,
Corresponding author.

Department of Community Medicine and General Practice, Medical faculty, Norwegian University of Science and Technology, N-7489 Trondheim, Norway.

Phone: ++ 47 73 59 01 37 (direct number)
Fax: ++ 47 73 59 87 89
E-mail: torstein.vik@ntnu.no

Magnus Hølmo Fasting.
First author